Author's response to reviews

Title: Use of television, videogames, and computer among children and adolescents in Italy

Authors:

Alessandro Patriarca (alessandro.patriarca@tin.it)
Gabriella Di Giuseppe (gabrydg@libero.it)
Luciana Albano (luciana.albano@unina2.it)
Paolo Marinelli (paolo.marinelli@unina2.it)
Italo F Angelillo (italof.angelillo@unina2.it)

Version: 5 Date: 11 May 2009

Author's response to reviews:

May 11, 2009

RE: “Use of television, videogames, and computer among children and adolescents in Italy”

Thank you very much for having accepted the manuscript “Use of television, videogames, and computer among children and adolescents in Italy” in BMC Public Health. We have made the following changes to the format of the manuscript so that online production is able to process the manuscript.

1. As suggested, the Table 2 is in only a single A4 portrait page.
2. As suggested, the Table 3 is in only a single A4 portrait page.
3. As suggested, we have removed the vertical lines from the Tables.
4. As suggested, we have reformulated the first sentence and we have made the other correction in the results section.
5. As suggested, we have used sentence case in the title and we have removed all unnecessary capitals.
6. As suggested, we have removed the authors’ qualifications.
7. As suggested, we have included the email addresses for all the authors on the title page.
8. As suggested, we have included the details of the previous work in the Acknowledgements section.
9. As suggested, we have removed the keywords.

Yours sincerely,
Italo F. Angelillo, DDS, MPH
Professor and Chairman